Author's response to reviews

Title: Influence of pain location and hand dominance on scapular kinematics and EMG activities: an exploratory study

Authors:

Yi-Fen Shih (yfshih@ym.edu.tw)
Yi-Hsuan Kao (kaoyihhsuan@gmail.com)

Version: 5 Date: 9 November 2011

Author's response to reviews:

Editorial Production Team, BMC Musculoskeletal Disorders
P.O. Box 35138, Philadelphia, PA 19128

09 Nov, 2011

Re: MS No. 1323692318535146

To Whom It May Concern:

We have formatted our manuscript entitled “Influence of pain location and hand dominance on scapular kinematics and EMG activities: an exploratory study” by Shih YF and Kao YH according to your suggestions:

1. Reference 19 and 20 are correctly cited;
2. Figure 1 is located in the Methods section;
3. Figures are cropped and the numbers of all figures are listed in the text;
4. Line numbers and underlining are removed.

Thank you for taking care of our revised manuscript.

Yours sincerely,

Yi-Fen Shih
Department of Physical Therapy and Assistive Technology
National Yang-Ming University
155, Li-Nong Street Section 2,
Taipei, Taiwan 112.
Tel: +886 2 28267340; fax +886 2 28201841